Author’s response to reviews

**Title:** The PstI/Rsal and Dral polymorphisms of CYP2E1 and head and neck cancer risk: a meta-analysis based on 21 case-control studies

**Authors:**
- Kefu Tang (tkf2006@163.com)
- Yang Li (yangli185@hotmail.com)
- Zhao Zhang (zuogouquan@hotmail.com)
- Yunmin Gu (guyunmin2@yahoo.com.cn)
- Yuyu Xiong (gracexyy84@yahoo.com.cn)
- Guoyin Feng (guoyingfeng@online.sh.cn)
- Lin He (helinhelin@gmail.com)
- Shengying Qin (chinsir@sjit.edu.cn)

**Version:** 9  **Date:** 6 July 2010

**Author’s response to reviews:**

we have corrected some typographical errors of the manuscript.